Author’s response to reviews

Title: Assessing the risk factors before pregnancy of preterm births in Iran: a population-based case-control study

Authors:

Maryam Soltani (maryam.soltani52@yahoo.com)
Hamid Reza Tabatabaee (tabatabaee@sums.ac.ir)
Shahin Saeidinejat (saeedish1@mums.ac.ir)
Marzieh Eslahi (marzieh.eslahi@gmail.com)
Halime Yaghoobi (Lemen_2012@yahoo.com)
Ehsan Mazloumi (Ehsan.mazloumi1@gmail.com)
Abdolhalim Rajabi (rajabiepid@gmail.com)
Ali Ghasemi (aria_201323@yahoo.com)
Naeimeh Keyghobadi (naeimehkeyghobady@yahoo.com)
Mostafa Enayatrad (mostafaenayatrad@gmail.com)
Abed Noori (Abed.noori@gmail.com)
Seyyed Aliasghar Hashemi (hashemishiz@gmail.com)
Fatemeh Zolfizadeh (fa.9712@yahoo.com)
Sepideh Mahdavi (S.mahdavi88@gmail.com)
Tannaz Valadbeigi (Tanazvaladbeigi@gmail.com)
Mahmoud Hajipour (m.hajipour.13@gmail.com)

Version: 1 Date: 28 Jan 2017

Author’s response to reviews:

Availability of data and materials:
The dataset(s) supporting the conclusions of this article is (are) available on request from the journals cites.

Ethical approval was granted by Shiraz university of Medical Sciences Research Ethics Committee (Ir.sums.rec.1394.f330). All participants provided written informed consent prior to interview; this included permission to use anonymised quotations in publications.